Author’s response to reviews

Title: Expanding the Role of Paralegals: Supporting Realization of the Right to Health for Vulnerable Communities

Authors:

Albert Wirya (awiry@lbhmasyarakat.org)

Ajeng Larasati (alarasati@lbhmasyarakat.org)

Sofia Gruskin (Gruskin@med.usc.edu)

Laura Ferguson (Laura.Ferguson@med.usc.edu)

Version: 2 Date: 27 Feb 2020

Author’s response to reviews:

Review: I am pleased to recommend publication subject to one very minor revision: on pp 21-22, line 9--the concluding line of the article--there is missing text. The sentence ends 'basic psychological.', so at least one word is missing. Once this change is made, I recommend publication.

Answer: Thank you for noticing the missing text. What we intended to write is 'basic counseling'. It has been revised in the draft I submitted.